Author's response to reviews

Title: Factor analysis of self-treatment in diabetes mellitus: a cross sectional study

Authors:

Negin Masoudi Alavi (alavi.negin@yahoo.com)
Leila Alami (alami233@yahoo.com)
Sedigheh Taefi (pooramini_en@yahoo.com)

Version: 2 Date: 6 June 2011

Author's response to reviews: see over
Dear Mr Silvestre

Thank you for your messages and recommendations. The following changes were made to the article “MS: 5869416548149202 - Factor analysis of self-treatment in diabetes mellitus: a cross sectional study”

1- The complementary information added about the questionnaire in method section.
2- A translated questionnaire was uploaded.
3- The object of the study was quoted in the end of the background.
4- The method section was revised and the exploratory factor analysis was described.
5- The result section was revised according to reviewers’ comments.
6- The discussion was revised and the limitations of the study were added to this section.
7- The tables were revised and the commented components were added to the tables.
8- All the changes in the article have been underlined.
9- A colleague of mine edited the article but I am not still sure about the quality of the language. In the case that you think it needs more editing, I am completely ready to send it for professional editing and pay the prices.

Sincerely Yours
Dr Negin Masoudi Alavi
Kashan University of Medical Science